Author’s response to reviews

Title: Local people's views on the evidence-based skilled-maternal-care in Mfuwe, Zambia: a qualitative study

Authors:

Choolwe Muzyamba (muzyamba@merit.unu.edu)

Version: 3 Date: 17 Mar 2019

Author’s response to reviews:

Dear Editor,

Thank you for your review and accepting our paper. Below you will find our responses to your specific comments.

Editor Comments:

1. In your “consent to publish” section, please confirm whether participants gave consent for direct quotes from their interviews to be published in this manuscript.

-We have included the statement

2. We note you have used Harvard Style referencing in your manuscript. The BMC Series uses Vancouver style. All references, including URLs, must be numbered consecutively, in square brackets, in the order in which they are cited in the text, followed by any in tables or legends. The reference numbers must be finalized and the reference list fully formatted before submission.

-We have corrected our reference style

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional
files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

-We have done so

Yours sincerely,

Dr. Choolwe Muzyamba